Author's response to reviews

Title: 'Pseudoneurological' symptoms, dissociation and stress-related psychopathology in healthy young adults

Authors:

Petr Bob (petrbob@netscape.net)
Petra Selesova (pselesova@gmail.com)
Jiri Raboch (raboch@nbox.cesnet.cz)
Lubomir Kukla (lubomir.kukla@tiscali.cz)

Version: 4 Date: 25 April 2013

Author's response to reviews: see over
Author's response to reviews
MS: 1527493397391127
Title: 'Pseudoneurological' symptoms, dissociation and stress-related psychopathology in healthy young adults

Authors:
Petr Bob, Petra Selesova, Jiri Raboch and Lubomir Kukla

Version: 2 Date: April 23, 2013

Author's response to reviews: see over
Referee 1:
Reviewer's report
Title: Somatoform dissociation, 'pseudoneurological' symptoms and stress-related psychopathology in healthy adolescent population
Version: 3
Date: 2 April 2013
Reviewer: Doris K Nilsson
Reviewer's report:

Minor but essential

I think the word confirm in the conclusion in the abstract is to strong" showed is better i think.
-- corrected

More outspoken limitations
Corrected in the discussion section: ...“Other significant limitation with respect to the results and their interpretation is that correlation does not mean causality and in fact the processes related to stress and specific changes in the nervous system have characteristic quality of circular causality which means that brain vulnerability or dysfunction most likely creates higher sensitivity to stress influences and on the other hand through conversion mechanism psychological stress may influence the brain and its sensitivity with respect to various insults and other environmental influences. “

clear references to the questionnaires used after each one.
-- corrected in the method section

Referee 2:
Reviewer's report

Title: Somatoform dissociation, 'pseudoneurological' symptoms and stress-related psychopathology in healthy adolescent population

Version: 2 no critical comment